Author's response to reviews

Title: Postprandial glucose-lowering effects of fermented red ginseng in subjects with impaired fasting glucose or type 2 diabetes: A randomized, double-blind, placebo-controlled clinical trial

Authors:

Mi-Ra Oh (omr@ctcf2.com)
Soo-Hyun Park (psh@ctcf2.com)
Sun-Young Kim (Kimsy@ctcr2.com)
Hyang-Im Back (hyangim100@gmail.com)
Min-Gul Kim (mqkim@jbctc.org)
Ji-Young Jeon (jyjeon@jbctc.org)
Ki-Chan Ha (omphalos@chonbuk.ac.kr)
Won-Taek Na (wonpharm@hanmail.net)
Youn-Soo Cha (cha8@jbnu.ac.kr)
Byung-Hyun Park (bhpark@jbnu.ac.kr)
Tae-sun Park (pts@jbnu.ac.kr)
Soo-Wan Chae (soowan@jbnu.ac.kr)

Version: 5
Date: 13 February 2014

Author's response to reviews:

Feb. 13, 2014
Managing Editor,
BMC Complem Altern Med

Manuscript Number 2298986108910509

Dear Editor,

Enclosed is the revised version of our manuscript, entitled “Postprandial glucose-lowering effects of fermented red ginseng in subjects with impaired fasting glucose or type 2 diabetes: A randomized, double-blind, placebo-controlled clinical trial” by Mi-Ra Oh et al. We feel deep thanks for reviewer’s kind comments. The manuscript has been revised according to the recommendations of reviewers. I hope that you will now find the manuscript satisfactory with following changes.

Sincerely,

Soo-Wan Chae, MD, PhD
Clinical Trial Center,
Chonbuk National University Hospital,
Jeonju, Jeonbuk, 561-756, Korea
Tel: 82-63-250-2348
Fax: 82-63-250-2349
Email:soowan@jbnu.ac.kr